Author's response to reviews

Title: Use and acceptance of long lasting insecticidal net screens for dengue prevention in Acapulco, Guerrero, Mexico

Authors:

- Catrin H Jones (catrinhuwsjones@gmail.com)
- David Benitez-Valladares (davidbeniva@gmail.com)
- Guillermo Guillermo-May (gqm.biol@gmail.com)
- Felipe Dzul-Manzanilla (fdzul@me.com)
- Azael Che-Mendoza (achemendoza_vectores@hotmail.com)
- Mario Barrera-Pérez (barreraperezma@gmail.com)
- Johannes Sommerfeld (sommerfeldj@who.int)
- Axel Kroeger (kroegera@who.int)
- Timothy O'Dempsey (T.Odempsey@liverpool.ac.uk)
- Pablo Manrique-Saide (pablo_manrique2000@hotmail.com)
- Celia Selem-Salas (ssalas@uady.mx)
- Juan Chablé-Santos (jcsantos@uady.mx)
- Anuar Medina-Barreiro (anuar116@hotmail.com)

Version: 7 Date: 16 July 2014

Author's response to reviews: see over
The changes that have been made are the following:

- The authors order has been changed being now Pablo Manrique-Saide the last autor instead of the second last.
- The abstract has been formatted according to BMC guidelines, using Background, Methods, Results and Conclusions sections only.